Author's response to reviews

Title: Long term life dissatisfaction and subsequent major depressive disorder and poor mental health

Authors:

Teemu Rissanen (teemu.rissanen@fimnet.fi)
Heimo Viinamäki (heimo.viinamaki@kuh.fi)
Kirsi Honkalampi (kirsi.honkalampi@kuopio.fi)
Soili M Lehto (soili.lehto@kuh.fi)
Jukka Hintikka (jukka.hintikka@kuh.fi)
Tarja Saharinen (tarja.saharinen@kuh.fi)
Heli Koivumaa-Honkanen (heli.koivumaa-honkanen@kuh.fi)

Version: 3 Date: 22 August 2011

Author's response to reviews:

Thank you for the important notices made concerning the article "Long term life dissatisfaction and subsequent major depressive disorder and poor mental health".

The changes and revisions to the format of the manuscript were made according to the request made by BMC Series Editorial Production Team 17th August 2011.

Sincerely,
Teemu Rissanen, corresponding author